Author's response to reviews

Title: Critical appraisal skills training for health care professionals: A randomized controlled trial [ISRCTN46272378]

Authors:

Rod R Taylor (r.s.taylor@bham.ac.uk)
Barnaby C Reeves (barney.reeves@lshtm.ac.uk)
Paul E Ewings (paul.ewings@tst.nhs.uk)
Rebecca J Taylor (r.j.taylor@bham.ac.uk)

Version: 4 Date: 2 December 2004

Author’s response to reviews:

Emma Parkin
Editorial Administrator, BMC Journals

Dear Emma

I have revised the manuscript as directed in your e-mail of 29th November 2004. However, I unable to undertake the cropping instructions on the figure. I am hoping that one of the editorial team may be able to do this on behalf!

Please contact me if you have any further queries.

Thank you for your assistance

Your sincerely

Dr Rod Taylor